Author's response to reviews

Title: Interaction and efficacy of Keigai-rengyo-to extract and acupuncture in male patients with acne vulgaris: A study protocol for a randomized controlled pilot trial

Authors:

Kyu Seok Kim (omdkskim@gmail.com)
Yoon-Bum Kim (kyb6838@hanafos.com)

Version: 5 Date: 12 March 2011

Author's response to reviews: see over
Cover letter

Dear Editor,

I am sorry for the inconvenience due to the revision of our paper.

You asked me to carry out the additional revisions listed as follows.

Major revisions:

Affiliations: You have 2 different affiliations on the title page (with different departments). Please indicate, with superscript numbers, in the author list and affiliations which author belongs to each affiliation.

Kyu Seok Kim and Yoon bum Kim are affiliated with both “Department of Ophthalmology & Otolaryngology & Dermatology, Hospital of Oriental Medicine, Kyung Hee University Medical Center, Seoul, Republic of Korea” and “Department of Oriental Medicine, College of Oriental Medicine, Kyung Hee University, Seoul, Republic of Korea”.

So we added and revised in the author list and affiliations which belongs to each affiliation with superscript numbers as follows.

Kyu Seok Kim\textsuperscript{a, b}, Yoon-Bum Kim\textsuperscript{a, b}\textsuperscript{*}

\textsuperscript{a}Department of Ophthalmology & Otolaryngology & Dermatology, Hospital of Oriental Medicine, Kyung Hee University Medical Center, Seoul, Republic of Korea

\textsuperscript{b}Department of Oriental Medicine, College of Oriental Medicine, Kyung Hee University, Seoul, Republic of Korea

Corresponding author’s address (Yoon-Bum Kim) was listed as only “Department of Oriental Medicine, College of Oriental Medicine, Kyung Hee University, 1 Hoegi-dong, Dongdaemun-gu, Seoul 130-701, Republic of Korea”. If you think we have to add another affiliation of corresponding author, please tell me by e-mail.

Thank you for your kind considerations in advance.
Sincerely yours,

Kyu Seok Kim O.M.D. Ph.D (I am not Mrs. but Mr. ^_^)